Author's response to reviews

Title: Self-perceived psychological stress is associated with parameters of adiposity and subclinical atherosclerosis disease in Mexican population. The GEA Study.

Authors:

Janinne Ortega-Montiel MD (janis_boly@hotmail.com)
Rosalinda Posadas-Sanchez MSc (rossy_posadas_s@yahoo.it)
Carlos Posadas-Romero MD (cposadasr@yahoo.com)
Wendy A Ocampo-Arcos BSc (mayte1190@yahoo.com.mx)
Aida X Medina-Urrutia MSc (aidaxm@yahoo.com)
Guillermo C Cardoso-Saldaña PhD (gccardosos@yahoo.com)
Esteban Jorge-Galarza MSc (esjoga@yahoo.com.mx)

Version: 5 Date: 28 July 2015

Author's response to reviews: see over
Frank Zimmermann-Viehoff
Reviewer

**Reply to comments by Reviewer 2**

**Reviewer's report: Minor essential revision:**

The language still shows considerable weaknesses (misspellings, expressions). In my opinion, language editing would be necessary before the manuscript can be published. All other points were adequately addressed.

As you recommended, we have reviewed the entire paper following the recommend structure and language editing. We have corrected the grammatical mistakes, misspellings and expressions on the manuscript. Some sentences have been change, without lose its principal meaning. A natural English speaker has made all these corrections.

I thank you for your attention and courtesy.

Sincerely,

Rosalinda Posadas-Sánchez, MSc.
Departamento de Endocrinología
Instituto Nacional de Cardiología “Ignacio Chávez”
Address: Juan Badiano No. 1 col. Sección XVI, Tlalpan,
C.P. 14080. Mexico, D.F., Mexico.
Tel: +(52)55732911 Ext. 1272 Fax:+(52)55734687
E-mail: rossy_posadas_s@yahoo.it